Author's response to reviews

Title: Evaluating the SOS Suicide Prevention Program: A Replication and Extension

Authors:

Robert H Aseltine Jr (aseltine@uchc.edu)
Amy James (ajames@uchc.edu)
Elizabeth A Schilling (eschilling@uchc.edu)
Jaime Glanovsky (jglanovsky@uchc.edu)

Version: 4 Date: 22 June 2007

Author's response to reviews:

As requested by the editor, the methods section now includes the name of the institutional entity providing human subjects approval for this study as well as an explanation that the anonymity of study participants allowed us to obtain a waiver of informed consent (i.e., 'passive' consent procedures were used).